Author's response to reviews

Title: A systematic review of the effectiveness of individual, community and societal level interventions at reducing socioeconomic inequalities in obesity amongst children

Authors:

Frances C Hillier-Brown (frances.hillier@durham.ac.uk)
Clare L Bambra (clare.bambra@durham.ac.uk)
Joanne-Marie Cairns-Nagi (j.m.cairns@durham.ac.uk)
Adetayo Kasim (a.s.kasim@durham.ac.uk)
Helen J Moore (helen.moore@durham.ac.uk)
Carolyn D Summerbell (carolyn.summerbell@durham.ac.uk)

Version: 3 Date: 18 July 2014

Author's response to reviews: see over
Dear Sir/Madam,

We would like to thank the reviewer for their helpful and constructive comments. We have revised our paper in light of these comments, and changes we have made are highlighted in yellow in the manuscript.

1689318881120414 - A systematic review of the effectiveness of individual, community and societal level interventions at reducing socioeconomic inequalities in obesity amongst children

Reviewer's report – Reviewer 2
R2.1 My only comment now relates to use of the term ‘obesity-related outcomes’. Although I had suggested this change in the text, using the term in the title may not be advisable as it could be misconstrued as referring to other health outcomes associated with obesity, such as diabetes.

We have amended the title of the manuscript to the following: A systematic review of the effectiveness of individual, community and societal level interventions at reducing socioeconomic inequalities in obesity amongst children

Thank you for your useful comments and hope that you find our revisions satisfactory. I look forward to hearing from you.

Yours faithfully,

Carolyn Summerbell

Principal of John Snow College  www.dur.ac.uk/johnsnow.college/

Professor of Human Nutrition, School of Medicine & Health, and Fellow of the Wolfson Research Institute  www.dur.ac.uk/school.health/centres/orb/